Author's response to reviews

Title: Treatment of unstable pelvic fractures through a less invasive ilioinguinal approach combined with minimally invasive posterior approach

Authors:

Lei Zhu (hailangzhulei@126.com)
Lu Wang (wanglu08@gmail.com)
Aimin Chen (aiminchen@aliyun.com)

Version: 2 Date: 12 November 2014

Author’s response to reviews:

Dear editor,

We have revised our manuscript to include line and page numbers. A statement of ethics permission and consent was included in the Methods section. We also added the Acknowledgements part in the manuscript.

Yours sincerely

Aimin Chen.